Developing primary care services for stroke survivors: the Improving Primary Care After Stroke (IPCAS) research programme

Ricky Mullis,¹ Maria Raisa Jessica Aquino,² Elizabeth Kreit,¹ Vicki Johnson,³ Julie Grant,¹ Emily Blatchford,¹ Mark Pilling,¹ Francesco Fusco¹ and Jonathan Mant^{1*} on behalf of the IPCAS investigators

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/ AYHW3622.

Primary conflicts of interest: Vicki Johnson declares that she is employed by the University Hospitals of Leicester NHS Trust within the Leicester Diabetes Centre (LDC), which receives not-for-profit income for DESMOND, a suite of self-management programmes for which LDC holds the Intellectual Property Rights. UHL also receives various grants to pay staff within LDC to carry out and implement various studies and self-management programmes. The Intellectual Property for MLAS is held by the University of Leicester on behalf of UHL. Jonathan Mant declares that he is a NIHR Senior Investigator, a subpanel chair for NIHR Programmes for Applied Health Research and a member of the NIHR CTU Standing Advisory Committee.

Published February 2024 DOI 10.3310/AYHW3622

Plain language summary

Developing primary care services for stroke survivors: the Improving Primary Care After Stroke (IPCAS) research programme

Programme Grants for Applied Research 2024; Vol. 12: No. 1

NIHR Journals Library www.journalslibrary.nihr.ac.uk

DOI: 10.3310/AYHW3622

¹Primary Care Unit, Department of Public Health and Primary Care, University of Cambridge, Cambridge, UK

²Population Health Sciences Institute, Newcastle University, Newcastle, UK ³University Hospitals Leicester NHS Trust, Leicester Diabetes Centre, Leicester, UK

^{*}Corresponding author jm677@medschl.cam.ac.uk

Plain language summary

What was the question?

People tell us that their longer-term problems after a stroke are not always well dealt with. We wanted to develop a new way for general practice to look after their patients who have had a stroke in such a way that these problems are better addressed.

What did we do?

We looked at the evidence that already exists in terms of what problems people face after their stroke. Then we spoke to people with stroke and their carers and to healthcare professionals to understand how general practice might be better able to support people. A multiprofessional group, including patients, met to discuss our findings and propose a new way of working. The approach included:

- review of patient needs by a practice nurse using a checklist adapted for use in general practice
- group-based self-management course
- · directory of relevant local services
- direct point of contact for people with stroke to use
- ways to improve communication between primary care and specialist staff
- training for practice nurses.

Over 1 year, we followed up patients with stroke in 23 practices who tried out this new model of care and compared what happened to them and how they felt with patients from 23 practices that had not introduced this new model.

What did we find?

We found that the commonest problems that people reported included fatigue, difficulties in thinking and mood. We found that the new model did not change the impact of stroke on patients who received it. Some participants indicated that the approach might have been more helpful earlier in their stroke journey.

What does this mean?

The services developed in this research are not relevant to all people after stroke in the longer term. Research is needed on how to address the specific poststroke problems that patients report.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

Programme Grants for Applied Research (PGfAR) was launched in 2013 and is indexed by Europe PMC, NCBI Bookshelf, DOAJ, Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and Scopus® (Elsevier, Amsterdam, Netherlands).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar.

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health and Care Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number PTC-RP-PG-0213-20001. The contractual start date was in April 2015. The final report began editorial review in April 2021 and was accepted for publication in October 2022. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the PGfAR programme or the Department of Health and Social Care.

Copyright © 2024 Mullis *et al.* This work was produced by Mullis *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

NIHR Journals Library Editors

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board, Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk